Author's response to reviews

Title: Four successful pregnancies in woman after Fontan palliation: Case report

Authors:

Rym M.D Gribaa (rym_gribaa@yahoo.fr)
Mehdi M.D Slim (mehdislim_fms@yahoo.fr)
Elies Prof. Neffati (elies_neffati@yahoo.fr)
Essia Prof. Boughzela (essiaboughzela@yahoo.fr)

Version: 2 Date: 29 June 2014

Author's response to reviews: see over
To,

The Editor

Sub: Submission of Manuscript for publication

Dear Sir,

We intend to publish an article entitled “Four successful pregnancies in woman after Fontan palliation: Case report” in your journal as a Case Report.

In our knowledge, it is the first case of four successful pregnancies with normal delivery of four healthy neonates in a woman with past history of Fontan palliation reported in the literature.

There is a routine tendency to discourage pregnancies after Fontan palliation. This case report and other few reports in the literature suggests that Women can successfully complete pregnancy after adequate Fontan palliation without important long-term sequelae.

Thanking you for your consideration,

Yours’ sincerely,

Signature

Corresponding contributor: Mehdi Slim, Service de cardiology, Hôpital Sahloul, Route de la ceinture, cité Sahloul, 4011, Hammam Sousse, Sousse, Tunisie

E-mail: mehdislim_fms@yahoo.fr